Author’s response to reviews

Title: How to estimate glomerular filtration rate in sub-Saharan Africa: design and methods of the African Research into Kidney Diseases (ARK) study

Authors:

Robert Kalyesubula (rkalyesubula@gmail.com)

June Fabian (june.fabian@mweb.co.za)

Wisdom Nakanga (wisdomnakanga@gmail.com)

Robert Newton (rob.newton@ecsg.york.ac.uk)

Billy Ssebunnya (billy.ssebunnya@mrcuganda.org)

Josephine Prynn (josephineprynn@gmail.com)

Jaya George (jaya.george@wits.ac.za)

Alisha Wade (alisha.wade@wits.ac.za)

Janet Seeley (janet.seeley@mrcuganda.org)

Dorothea Nitsch (dorothea.nitsch@lshtm.ac.uk)

Christian Hansen (Christian.Hansen@lshtm.ac.uk)

Moffat Nyirenda (moffat.nyirenda@lshtm.ac.uk)

Liam Smeeth (liam.smeeth@lshtm.ac.uk)

Saraladevi Naicker (saraladevi.naicker@wits.ac.za)

Amelia Crampin (mia.crampin@lshtm.ac.uk)

Laurie Tomlinson (laurie.tomlinson@lshtm.ac.uk)

Version: 3 Date: 28 Dec 2019

Author’s response to reviews:

. Ethics approval and consent to participate
Within this section please specify whether informed consent to participate will be obtained in written or verbal format and if verbal, please provide a reason why verbal consent to participate was used rather than written consent.

This has been included as written informed consent in the ethics sections.

2. Formatting

Please remove the Conclusions section from the main text of this manuscript according to journal guidelines:

https://bmcnephrol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/study-protocol and you may incorporate the relevant contents into the Discussion section if appropriate.

the conclusion has been removed and incorporated into the discussion section.

3. Source of figure 1

Please clarify whether figure 1 is your own or taken from another source.

If this was created by you, please state so in the figure caption. If this was not created by you, please state the source in the figure caption. Please note that for the publication of any copyrighted images, BMC requires that permission to publish these has been obtained from the copyright holder. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it. If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

we have included a caption to show that the map was developed by Robert Kalyesubula and illustrated by Helmut Kraus as original documents for this manuscript.

4. Clean copy

On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please remove ‘BMC NEPHROLOGY 2019 RESPONSES 18 Oct.docx’ from the file inventory as it is no longer required at this stage.